Author's response to reviews

Title: Using qualitative mixed methods to study small health care organizations while ensuring trustworthiness and authenticity

Authors:

Christine B Phillips (christine.phillips@anu.edu.au)
Kathryn M Dwan (kathryn.dwan@anu.edu.au)
Julie Hepworth (julie.hepworth@qut.edu.au)
Christopher M Pearce (chris_pearce@mac.com)
Sally Hall (sally.hall@anu.edu.au)

Version: 2 Date: 22 May 2014

Author's response to reviews:

Dear Dr Morrey

"Using qualitative mixed methods to study small health care organizations while ensuring trustworthiness and authenticity"

Please accept our apologies for the earlier incomplete submission. Omitting the title page was a regrettable oversight. In being rectified we have addressed five of the six requirements.

• Title page
• Competing interests
• Authors’ contribution
• Acknowledgements
• Ethical clearance

Renaming the sections of the manuscript so they read as Background, Methods, Results, Discussion, Conclusions, and List of abbreviations addresses the final requirement listed under the Manuscript body.

Yours sincerely

Kathryn Dwan